Author’s response to reviews

Title: Studying medical communication with video vignettes: how variations in video-vignette introduction format and camera focus influence analogue patients' engagement. A randomized study.

Authors:

Leonie Visser (n.c.visser@amc.uva.nl)
Nadine Bol (n.bol@uva.nl)
Marij Hillen (m.a.hillen@amc.uva.nl)
Mathilde Verdam (m.g.e.verdam@amc.uva.nl)
Hanneke de Haes (j.c.dehaes@amc.uva.nl)
Julia van Weert (j.c.m.vanweert@uva.nl)
Ellen Smets (e.m.smets@amc.uva.nl)

Version: 3 Date: 11 Dec 2017

Author’s response to reviews:

Dear Editor,

Using the Biomed Central form, we have obtained consent from the male actor that plays the male video patient in our video vignettes used in this study. He is the person in the image used as part of Figure 1. This consent form is available for review upon request.

We have now added the following section to the declarations of the manuscript:

Consent for publication

Written informed consent for publication of the image that is part of Figure 1 was obtained from the professional actor that acted as the male video patient in our video vignettes. A copy of the consent form is available for review by the Editor of this journal.